Author's Response To Reviewer Comments

Dear Scott,

We have added the protocol into protocols.io and integrated the DOI (dx.doi.org/10.17504/protocols.io.ka9csh6) in the revised article. We also have checked and completed the spreadsheet from GigaDB to complete the related GigaDB dataset page. Thank you for all your help and looking forward to hearing from you soon.

Sincerely yours, Xin Zhou